Author's response to reviews

**Title:** Lung cancer mortality in towns near paper, pulp and board industries in Spain

**Authors:**

Susana Monge-Corella (smonge@isciii.es)  
Javier Garcia-Perez (jgarcia@isciii.es)  
Nuria Aragones (naragones@isciii.es)  
Marina Pollan (mpollan@isciii.es)  
Beatriz Perez-Gomez (bperez@isciii.es)  
Gonzalo Lopez-Abente (glabente@isciii.es)

**Version:** 2  **Date:** 14 March 2008

**Author's response to reviews:**

Graphics are improved.